Response to peer reviewers’ comments
Thank you for acknowledging the content of the article. The regulation of chemotherapeutic sensitivity via lncRNAs in autophagy process have been given in a figurative form.

Response to Science Editor’ comments
(1) "Country/Territory of origin: China" submitted by the system is correct.
(2) We have sought the help of professional organizations for language editing and have obtained the appropriate certificates to prove that.
(3) The “Running Title” has been added: Yu JM et al. LncRNAs regulate autophagy and chemoresistance.
(4) “Author contributions” has changed: Yu JM, Sun CQ and Liu LX contributed to this paper; Yu JM, Sun CQ, Xu HH, Jiang YL and Liu LX designed the overall concept and outline of the manuscript; Yu JM, Sun CQ, Jiang XY, Ni SQ, Zhao TY and Liu LX contributed to the discussion and design of the manuscript; Yu JM and Sun CQ contributed to the writing, and editing the manuscript, illustrations, and review of literature; and all authors read and approved the final manuscript.
(5) The Core Tip has made some changes: Abbreviations have been defined upon first appearance in the Core Tip (Long non-coding RNAs (LncRNAs) and colorectal cancer (CRC)).
(6) The format of in-text citation of references is [References Number]. There is no space between “[ ]” and the preceding word. The reference number has been placed immediately after the author(s) of the reference. The order and total number of references cited have been verified.
(7) The WJGO article which this editorial discussed has been listed in the references list.
(8) Main text’s first level subtitles have been all capitalized, bolded and underlined.